Author's response to reviews

Title: Alcohol consumption is associated with lower self-reported disease activity and better health-related quality of life in female rheumatoid arthritis patients in Sweden. Data from BARFOT, a multicenter study on early RA.

Authors:

  Stefan Bergman (stefan.bergman@spenshult.se)
  Sofia Symeonidou (sofia.symeonidou@axessmedica.se)
  Maria L Andersson (maria.andersson@spenshult.se)
  Maria K Söderlin (maria.soderlin@spenshult.se)

Version: 3  Date: 25 May 2013

Author's response to reviews:

According to the suggestions from the editor we have improved the English and checked the style of the manuscript. The titlepage was changed according to the checklist. No other revisions were suggested of performed.